Author's response to reviews

Title: The role of social networks in the development of overweight and obesity among adults: a scoping review

Authors:

Katie Powell (k.powell@sheffield.ac.uk)
John Wilcox (John.Wilcox@nottinghamcity.gov.uk)
Angie Clonan (angie.clonan@sheffield.ac.uk)
Paul Bissell (p.bissell@sheffield.ac.uk)
Louise Preston (l.r.preston@sheffield.ac.uk)
Marion Peacock (m.peacock@lancaster.ac.uk)
Michelle Holdsworth (michelle.holdsworth@sheffield.ac.uk)

Version: 2
Date: 6 March 2015

Author's response to reviews:

The following changes have been made to the manuscript.

1) Line numbers have been added

2) The Competing Interests, Authors’ Contributions and Acknowledgement sections have been moved to after the Conclusions section.